INFRARED INTERPRETER’S DAILY LOG

	Incident Name:

Basin Complex
	IR Interpreter(s):

Rob Aiken
	Local Dispatch Phone:

805-938-9142
	Interpreted Size:

90580  acres
Growth last period:

11405 acres

	Flight Time:
2112 MDT
Flight Date:
20080708
	Interpreter(s) location:
North Ops, Redding, CA
Interpreter(s) Phone:

303-250-0699
	Regional Coordinator:

Melinda McGann
Regional Coord. Phone:
720-560-6971
	National Coordinator:

     
National Coord. Phone:
208-387-5381

	Ordered By:

Mark Steele
	Order Number:

     
	Aircraft/Scanner System:

144Z
	Pilots/Techs:

 Dan Johnson / Jessie Barain  

	IRIN Comments on imagery:

Good, minor marine haze along coast
	Weather at time of flight:

     
	Flight Objective:

fire perimeters

	Date and Time Imagery Received by Interpreter:

20080708 2300
	Type of media for final product:
shapefiles, .kml's
Electronic file sent to:
ftp://ftp.nifc.gov/Incident_Specific_Data/CALIF_S/2008/2008_Basin_Complex_LPF/GIS/IR/20080709

	Date and Time Imagery Delivered to Incident:

20080709  0530
	

	Comments /notes on tonight’s mission and this interpretation:
     The largest areas of growth occurred in the north.  Additional growth at the south east corner of the fire near the Indian complex.

Hot areas:  

 At the north end fire has converged from north and south in the Little Sur Drainage.  In the northeast corner heat perimeter has passed over the Ventana trail and Little Pines Camp.  Growth was observed down the east flank of Uncle Sam Mountain and heat is within approximately 0.5 miles of an area identified as Hiding Canyon Camp.
At the south end of the fire,  beginning just south of Lost Valley Creek and extending approximately 4.5 miles to the southeast is an area of active growth with intense heat.  The actual heat perimeter may extend further to the south as the intense heat area runs to the edge of the imagery.  IR flight crew informed that the flight was cut short by Air Traffic Control but that they felt confident that they had recorded to the extent of the heat.
Still a few heat sources seen on the Indians Complex.  The heat perimeter has not changed since last night.
 I have included kml files of the IR interpretation. 



